Author's response to reviews

Title: Evaluation of a safer male circumcision training programme for traditional surgeons and nurses in the Eastern Cape, South Africa

Authors:

Karl Peltzer (kpeltzer@hsrc.ac.za)
Ayanda Nqeketo (Anqeketo@hsrc.ac.za)
George Petros (GPteros@hsrc.ac.za)
Xola Kanta (Xkanta@telcom.sa)

Version: 2 Date: 14 December 2007

Author's response to reviews:

again revisions were made, reduce introduction